Notes for the July 6, 2016, CERES Processing Oversight Board Meeting

Meeting began at 1:05 p.m. in the conference room in B2102 R116.

Meeting Notes

No comments on the notes for the previous meeting.

Announcements

- July 20th CPOB meeting will be held in either B1250 R116 or 1EP Suite 215
- Fall CERES STM Oct. 18 21, 2016, European Centre for Median-Range Weather Forecasts (ECMWF), Shinfield Park, Reading, England

SCCR Discussion (SCCRs that have been submitted or updated since the last meeting)

ERBE-like 1145 – Disapproved and replaced by SCCRs 1151 and 1152. CATALYST 1146 – Updating to reduce the lag in displaying job lists in the Operator's Console. TISA Averaging 1147 – To resolve an issue found in production for September 2003. SARB 1148 – Edition2 MATCH files for January through April 2016. CATALYST 1149 – Database update to account for new GGEO files. SARB 1150 – Corrected Edition4 MATCH files for January through June 2013. ERBE-like 1151 – Delivering code to CM repository. ERBE-like 1152 – Delivering files to CERESlib (SCCR 1133). Inversion 1153 – Updated SORCE file through June 2016.

There were no issues identified, so these SCCRs will be approved.

CATALYST Status

Continue to test SCCR 1101.

The PGEs associated with the CATALYST modules delivered under SCCR 1168 have been running smoothly in production. They are running 33% faster than they did when they were run manually.

A CATALYST 2.0 training session will be held July 13, 2016.

Production Processing Status

No updates to the posted report.

System Status

The new blades are not being used by the DMT very much yet.

Status of Open SCCRs

In development:

- 1150 Will be delivered to dropbox today.
- 1153 July 8 target delivery date.
- 1121 New target delivery date: July 22.
- 1134 New target delivery date: July 29.
- 1135 No update.

1124 – Need JCT3B data to test code.

- 1139 No update.
- 1125 No update.
- 1140 No update.
- 1133 No update.
- 1144 No update.
- 1130 Need to decide if this is needed pre-launch.
- 1141 No update.

In CM:

- 1115 Hope to release by next week.
- 1101 Hope to release by next week.
- 1132 In CM queue.

In SIT:

- 1122 On hold until 1121 is promoted.
- 1108 Wrapper testing is complete. Ready to promote to the PPE.
- 1129 Ready to promote to the PPE.
- 1147 In ASDC queue.

Additional Items

The Promotion Priorities were reviewed.

Meeting ended at 2:09 p.m.